Author's response to reviews

Title: Assessing the efficacy of a modified assertive community-based treatment programme in a developing country.

Authors:

Ulla A Botha (ulla@sun.ac.za)
Liezl Koen (liezlk@sun.ac.za)
John A Joska (john.joska@uct.ac.za)
Linda Hering (Lhering@pgwc.gov.za)
Piet P Oosthuizen (pieto@sun.ac.za)

Version: 5 Date: 8 September 2010

Author's response to reviews:

Dear Sir,

we have made the major and some of the minor revisions requested. We are uncertain about the requested changes to the tables and would be grateful if you could do these for us.

regards,

Ulla Botha